Author's response to reviews

Title: Masked Volume Wise Principal Component Analysis Improves Signal Detection, Separation and Analysis of Small Adrenocortical Tumors in Dynamic [11C]-Metomidate Positron Emission Tomography

Authors:

Pasha Razifar (pasha.razifar@ge.com)
Joakim Hennings (joakim.hennings@surgsci.uu.se)
Azita Monazzam (azita.monazzam@ge.com)
Per Hellman (per.hellman@surgsci.uu.se)
Bengt Langstrom (bengt.langstrom@biorg.uu.se)
Anders Sundin (anders.sundin@ki.se)

Version: 3 Date: 14 March 2009

Author's response to reviews:

Dear Editor,

The authors of the manuscript have added the following sentences in body text of the manuscript and has uploaded a new version of the manuscript as you wished for. We hope that it is satisfactory enough for publication.

"The study was approved by the local ethics committee (No 02-262) and was performed in compliance with the Helsinki Declaration. Inclusion into the study was by informed consent."